Author’s response to reviews

Title: An integrated study of human and animal infectious disease in the Lake Victoria Crescent small-holder crop-livestock production system, Kenya

Authors:

Eric Fevre (Eric.Fevre@liverpool.ac.uk)
William de Glanville (Will.deGlanville@glasgow.ac.uk)
Lian Thomas (Lian.Thomas@zoonotic-diseases.org)
Elizabeth Cook (e.cook@cgiar.org)
Samuel Kariuki (skariuki@kemri.org)
Claire Wamae (nwamae@kemri.org)

Version: 2 Date: 14 Jun 2017

Author’s response to reviews:

1. We include a 'clean' manuscript with track changes accepted.

2. We have uploaded data to a repository, and included the link in the "Availability of data and materials" in the manuscript.